Author's response to reviews

Title: Problematic computer gaming, console-gaming, and internet use among adolescents: new measurement tool and association with time use

Authors:

Mette Rasmussen (mera@niph.dk)
Trine P Pedersen (tppe@niph.dk)
Pernille Bendtsen (pebe@niph.dk)
Katrine R Madsen (krm@niph.dk)
Charlotte R Meilstrup (chme@niph.dk)
Line Nielsen (linn@niph.dk)
Bjørn E Holstein (bho@niph.dk)

Version: 2 Date: 25 April 2013

Author's response to reviews:

Dear editor,

Thank you for the possibility to re-submit the manuscript with new information about ethical and data protection issues.

Kind regards, on behalf of the authors

Bjørn Holstein